### POLITECNICO DI TORINO Repository ISTITUZIONALE

Fundamental Concepts of Bipolar and High-Density Surface EMG Understanding and Teaching for Clinical, Occupational, and Sport Applications: Origin, Detection, and Main

Original

Fundamental Concepts of Bipolar and High-Density Surface EMG Understanding and Teaching for Clinical, Occupational, and Sport Applications: Origin, Detection, and Main Errors / Campanini, Isabella; Merlo, Andrea; Disselhorst-Klug, Catherine; Mesin, Luca; Muceli, Silvia; Merletti, Roberto. - In: SENSORS. - ISSN 1424-8220. - ELETTRONICO. - 22:11(2022), p. 4150. [10.3390/s22114150]

Availability: This version is available at: 11583/2966764 since: 2022-06-13T07:57:26Z

Publisher: MDPI

Published DOI:10.3390/s22114150

Terms of use:

This article is made available under terms and conditions as specified in the corresponding bibliographic description in the repository

Publisher copyright

(Article begins on next page)





## Review Fundamental Concepts of Bipolar and High-Density Surface EMG Understanding and Teaching for Clinical, Occupational, and Sport Applications: Origin, Detection, and Main Errors

Isabella Campanini <sup>1</sup>, Andrea Merlo <sup>1,2</sup>, Catherine Disselhorst-Klug <sup>3</sup>, Luca Mesin <sup>4</sup>, Silvia Muceli <sup>5</sup> and Roberto Merletti 6,\*

- 1 LAM-Motion Analysis Laboratory, Neuromotor and Rehabilitation Department, S. Sebastiano Hospital, Azienda USL-IRCCS di Reggio Emilia, Via Circondaria 29, 42015 Correggio, Italy; isabella.campanini@ausl.re.it (I.C.); andrea.merlo@ausl.re.it or ingmerlo@me.com (A.M.)
- 2 Merlo Bioengineering, 43121 Parma, Italy

5

- 3 Department of Rehabilitation & Prevention Engineering, Institute of Applied Medical Engineering, RWTH Aachen University, Pauwelsstr. 20, 52074 Aachen, Germany; disselhorst-klug@ame.rwth-aachen.de
- 4 Mathematical Biology and Physiology Group, Department of Electronics and Telecommunications, Politecnico di Torino, Corso Duca degli Abruzzi 24, 10129 Turin, Italy; luca.mesin@polito.it
- Division of Signal Processing and Biomedical Engineering, Department of Electrical Engineering,
- Chalmers University of Technology, Hörsalsvägen 11, 41296 Gothenburg, Sweden; muceli@chalmers.se Laboratory for Engineering of the Neuromuscular System (LISiN), Department of Electronics and
- Telecommunications, Politecnico di Torino, Corso Duca degli Abruzzi 24, 10129 Turin, Italy Correspondence: roberto@robertomerletti.it

Abstract: Surface electromyography (sEMG) has been the subject of thousands of scientific articles, but many barriers limit its clinical applications. Previous work has indicated that the lack of time, competence, training, and teaching is the main barrier to the clinical application of sEMG. This work follows up and presents a number of analogies, metaphors, and simulations using physical and mathematical models that provide tools for teaching sEMG detection by means of electrode pairs (1D signals) and electrode grids (2D and 3D signals). The basic mechanisms of sEMG generation are summarized and the features of the sensing system (electrode location, size, interelectrode distance, crosstalk, etc.) are illustrated (mostly by animations) with examples that teachers can use. The most common, as well as some potential, applications are illustrated in the areas of signal presentation, gait analysis, the optimal injection of botulinum toxin, neurorehabilitation, ergonomics, obstetrics, occupational medicine, and sport sciences. The work is primarily focused on correct sEMG detection and on crosstalk. Issues related to the clinical transfer of innovations are also discussed, as well as the need for training new clinical and/or technical operators in the field of sEMG.

Keywords: surface electromyography; surface EMG signal; high-density surface EMG; teaching; electrodes; crosstalk; volume conductor; conduction velocity; modeling

#### 1. Introduction

Bioelectric signals are the results of the activity of organs that are made of excitable cells, such as the brain, the heart or a muscle. Electrical currents flow through the membranes of these cells and generate action potentials that travel along axons or muscle fibers, triggering other events within or between the cells (e.g., the release of neurotransmitters and the contraction of a muscle fiber). The electric fields generated by these action potentials produce a distribution of electric potential in space, on the skin, that evolves over time, as with a movie.

Our eyes cannot see these distributions of voltage or their time evolution on the scalp (electroencephalogram, EEG), on the chest (electrocardiogram, ECG) or on a muscle (surface electromyogram, sEMG), but instruments can and they can visualize them as



Citation: Campanini, I.; Merlo, A.; Disselhorst-Klug, C.; Mesin, L.; Muceli, S.; Merletti, R. Fundamental Concepts of Bipolar and High-Density Surface EMG Understanding and Teaching for Clinical, Occupational, and Sport Applications: Origin, Detection, and Main Errors. Sensors 2022, 22, 4150. https://doi.org/10.3390/s22114150

Academic Editors: Francesco Carlo Morabito and Chang-Hwan Im

Received: 30 April 2022 Accepted: 26 May 2022 Published: 30 May 2022

Publisher's Note: MDPI stays neutral with regard to jurisdictional claims in published maps and institutional affiliations



Copyright: © 2022 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/).

colored images that we can see and interpret in the same way as thermal images describing temperature distributions over surfaces. To obtain images of voltage distribution, grids of contact points (electrodes) are applied to the chest, scalp or on a muscle, as indicated in Figures 1 and 2. Alternatively, an individual electrode (or a single electrode pair) detects local monopolar voltage with respect to a reference (or a differential voltage), providing a time-varying trace (as with ECG).



**Figure 1.** Maps of bioelectric signals: (**a**) example of an instantaneous surface EEG map obtained by a grid of electrodes on the scalp (reproduced from Figure 2 of [1]); (**b**) example of an instantaneous surface ECG map obtained by the simulation of a grid of electrodes on the torso (reproduced from Figure 4 of [2]).



**Figure 2.** Example of motor unit action potential (MUAP) generation and propagation: electrode grids and maps of the right and left propagating potentials on the skin. Modified from (https://www.robertomerletti.it/en/emg/material/teaching/module5 accessed 29 April 2022, This date applies to all websites and URLs). (See animation in Supplementary Materials, Sup1).

Bioelectric signals and images provide an extraordinary opportunity to discover and quantify a wealth of physiological and pathological phenomena that are associated with the organs that generate them. In addition, they provide valuable information to clinicians for functional diagnosis, treatment planning, and the evaluation of the effectiveness of interventions and patient recovery. The role of EEG mapping for studying brain rhythms, the spatial distribution of evoked potentials, epilepsy foci, etc., and the role of ECG mapping in studying atrio-ventricular conduction problems, arrhythmias, etc., are both well known to specialists. The role of sEMG in clinical medicine is less recognized, despite the potential benefits it may offer, mostly because of limited academic teaching [3]. Surface EMG signals are very rich in information about the muscle and neuromuscular system control strategies and can be easily detected by a number of different electrode configurations, with each having advantages, disadvantages, and specific applications.

The purpose of this work is to illustrate in a simple way several fundamental concepts that should be understood and recommendations that should be followed concerning the detection modalities of sEMG (electrode size, interelectrode distance (IED), electrode locations, etc.) in order to (a) be able to extract the information of interest and (b) apply it clinically. The applications of interest range from gait analysis in neurorehabilitation (Section 4.5) to the location of the innervation zone (IZ) for the targeted injection of botulinum toxin (Section 4.3) [4,5], from the estimation of load sharing among muscles [6] to the estimation of muscle fiber conduction velocity to monitor channelopathies and Cushing's syndrome (Section 2.2) [7–10], from monitor motor unit (MU) rate modulation and involuntary muscle activity in acute stroke [11–13] to study cramps [14], from comparing the ergonomics of chairs (Section 4.6) [15] to detecting when and which muscles are active and which are not [16], along with many other applications in sport and obstetric and rehabilitation medicine.

This work does not aim to present or discuss advances in sEMG technology. Its goal is to illustrate the current technology, in terms of teaching material, tutorials, and consensus papers, concerning the detection and interpretation of sEMG. This work is aimed at teachers, clinical tutors, and students with the objectives of:

- (a) overcoming some of the barriers that currently limit the widespread application of sEMG in the clinical world [17,18];
- (b) responding to the persistent demand for the simplification of teaching these concepts;
- (c) making users aware of the tutorials and teaching materials that are available.

Teaching tools and models are illustrated for this purpose. Some of them were derived from published tutorials [19,20] or modified from the free teaching website https://www.robertomerletti.it/en/emg/material/teaching, where further details can be found. They are presented with the intention of promoting teaching and outlining the improper or incorrect acquisition/interpretation of sEMG.

This paper is structured in six sections with the following purposes: Section 1 introduces the work and its motivations; Section 2 reviews a few basic physiological and technical concepts; Section 3 addresses the issue of sEMG detection using the basic spatial filters; Section 4 addresses the teaching of some clinical applications, focusing on the incorrect use of bipolar and high-density sEMG (HDsEMG); Sections 5 and 6 discuss the current situation and present conclusions.

#### 2. Teaching Basic Physiological and Technical Concepts

Fundamental physiological concepts about sEMG signals, events, and phenomena are reviewed here, together with sEMG modeling techniques that are suitable for teaching, as well as demonstrations.

#### 2.1. The Concept of Motor Unit Action Potential (MUAP)

The sEMG voltage distribution on the skin is the sum of the propagating MU action potentials (MUAPs) produced by the recruited MUs. The MU is the basic muscle building block, consisting of one motor neuron and all of the muscle fibers that it innervates. Every time an axonal action potential reaches the neuromuscular junction (NMJ), acetylcholine is released and a muscle fiber action potential is triggered. Figures 2 and 3 illustrate the concept of MU and the concept of propagating MUAP as a voltage distribution moving on the skin under a fixed grid of electrodes or under a pair of electrodes that are sampling it.

As indicated in Figure 3, each MUAP is the sum of the propagating surface action potentials produced by the single fibers of each MU and the sEMG is the sum of these MUAPs. Force, as well as the features of the sEMG, depends on many factors, including the number of recruited MUs and the discharge rate of each MU, which are both controlled by

the central nervous system [21]. The propagation (or conduction) velocity (CV) of the action potentials from the NMJ (or end plate) to the tendon endings depends on the dynamics of the fiber membrane ionic channels.



**Figure 3.** Schematic representation of the fibers of two motor units (MU1 and MU2) that are innervated by two motoneurons (MN1 and MN2) in two innervation zones (IZ1 and IZ2): (**a**) the summation of the skin contributions of the propagating action potentials, as detected by a differential amplifier (bipolar or single differential detection), are depicted as MUAP1 and MUAP2 (observe that the contributions of deeper muscle fibers are smaller than those of the superficial fibers of each MU); (**b**) the interferential signal, which is the summation of the MUAP trains of many active MUs. Modified from Figure 5.1 of [22] with permission.

A fundamental difference between sEMG and ECG or EEG is that electrodes placed on the skin, parallel to the muscle fibers, detect the MUAPs that are propagating bidirectionally from the IZs of the MUs to the muscle–tendon junctions. Since the fibers of an MU are of the same histological type and have about the same diameter, their action potentials travel at the same velocity (CV) along the fibers and their contributions add up, on the skin, as MUAPs that provide a "signature" or "fingerprint" for each MU.

These MUAPs are the basic building blocks of the sEMG. Understanding sEMG implies the understanding of its generation mechanisms. For this reason, our explanations of the physiological concepts focus mostly on single fiber action potentials and/or on single MUAPs whose sum is the sEMG. Figure 3 shows how the contributions of the single fiber action potentials of the MUs add up on the skin to form the MUAPs and how the MUAP contributions add up on the skin to form the sEMG that is detected by one electrode pair.

It should be noted that with progressively stronger voluntary contractions of a muscle, the information about the individual MUAPs becomes progressively more difficult to extract [23].

Although this may seem a simple and straightforward mechanism that is easy to describe and to teach students, a much deeper analysis is needed to fully understand its details. This is necessary to properly pose and answer the following questions: "What information about the MU properties and their central control strategy can be obtained from the sEMG?" and "When can this information be useful?".

# 2.2. *The Concept of Propagating Action Potentials and Their Velocity and Diffusion* 2.2.1. Conduction Velocity

For teaching purposes, simple metaphors and examples may be appropriate. One such example is provided in Figure 4. The water channel represents a muscle fiber, the falling stone represents the activation of the NMJ, the wave represents the propagating action potential, the floats represent the electrodes, and their displacement with respect to the resting water level represents the action potential that is detected by each electrode, considering that the wave is as wide as the water channel.



# **Figure 4.** Example of a bidirectional wave generated by dropping a stone in a narrow water channel (representing a single fiber). The floats represent a series of electrodes detecting the wave in subsequent times and in two directions. One pair of floats at equal distances from the stone detect identical waves and their difference is zero. Knowing the distance between floats and the delay between the waves they detect allows for the estimation of the wave propagation velocity. (See animation in Supplementary Materials, Sup2). Courtesy of Dr. Alberto Botter.

In practice, each muscle fiber can be considered as a line and an action potential moving along it can be approximated by three point-like current sources (a tripole), as described in detail in [24–26] and on https://www.robertomerletti.it/en/emg/material/teaching/module9. As is indicated also in Figure 10, the central arrow of the tripole represents positive ions entering the fiber through the open channels and the two lateral arrows represent the positive ions exiting the fiber or negative ions entering it. For the ease of representation and calculation, these ionic currents are modeled as concentrated at three specific points (the tripole).

Figure 5 shows the two MUAPs propagating to the right and left of the IZ of a threefiber motor unit. The volume conductor is layered (muscle, subcutaneous tissues, and skin). The red bars represent the single-fiber depolarized regions. Three detection points (P1, P2, and P3) are placed on the skin to detect the MUAP traveling to the right. The animation (Supplementary Materials, Sup3) shows the concept of propagation in time and space, as well as how the MUAP CV can be estimated by placing a series of electrodes along the



muscle fiber direction. The electrode pair P4 and P5 detects a differential signal near zero because of the symmetrical position of P4 and P5 with respect to the IZ.

**Figure 5.** Schematic drawing of the right and left propagation of a motor unit action potential (MUAP) in time and space and the estimation of muscle fiber conduction velocity. When the distance between P1 and P3 is 20 mm and the delay is 5 ms, the muscle fiber conduction velocity is 20/5 = 4 mm/ms = 4 m/s. P1, P2, and P3 must be on the same side of the innervation zone. The signals detected by P1, P2, and P3 are called "monopolar". The signals detected between P1–P2, P2–P3, and P1–P3 are called "bipolar" or "single differential (SD)". The SD signal detected between P4 and P5 is near zero because of the symmetrical position of P4 and P5 with respect to the innervation zone. Modified from https://www.robertomerletti.it/en/emg/material/teaching/module5. (See animation in Supplementary Materials, Sup3).

A single fiber action potential, or MUAP, produces a two-dimensional wave of electric potential over the skin that moves along the muscle (Figures 2 and 5). To summarize, MUAPs produce waves of monopolar potential across the skin over a muscle. Similar to water waves, these potentials travel on the skin from the region above the NMJ toward the tendons. During travel, they are detected by the electrodes. Electrode pairs are connected to a probe that contains a circuit, called a differential amplifier, that computes the difference between the two potentials that are present, instant by instant, below the two electrodes. Since the sEMG potentials are traveling along muscle fibers, the wave of potential of each MU reaches the electrodes at different times. This process is illustrated in Figure 5 and Figures 7–9.

#### 2.2.2. The Concepts of Volume Conductor and Crosstalk

We now look at the skin from above as the surface of a liquid containing light sources. For simplicity, we now consider a point-like light source that is at some depth in this liquid, which causes a halo to appear on the surface. In the same way, the source of an electric field at some depth below the skin produces a voltage distribution on the skin surface due to diffusion in the "conductive volume" (or "volume conductor") of the tissue, as indicated in Figure 6a. This effect is known as the "spatial low-pass filter effect", which is introduced by the volume conductor and means that MUs that have a greater distance from the detection system generate smaller, but spatially more extended, potential distributions.

Two point-like light sources (one blue and one red) produce the two partially overlapping halos, as represented in Figure 6b,c. The same happens for MUAPs belonging to different muscles: their surface representations can reach nearby muscles. This phenomenon is called "crosstalk", i.e., the presence of potentials on a target muscle that are produced by a different muscle. This may happen because of the blurring effect that is produced by the subcutaneous tissues and by a further phenomenon called the end-offiber (EOF) effect, which is discussed later. The following section explains the detection modalities that have been developed to partially compensate for this blurring effect, thus providing "spatial selectivity" to the sEMG measurement, which is often a pre-requisite for clinical application.



**Figure 6.** Overlapping of the distributions of surface intensities: (**a**) distribution of intensity generated by a point source of light (or a source of electric potential) at depth h below the surface; (**b**) partially overlapping distributions of light intensities generated by two sources at depth h; (**c**) profiles of partially overlapping light intensities on the surface and along the line L. The same considerations apply to point sources (S<sub>1</sub> and S<sub>2</sub>) of electric potential whose surface maps extend along x and y, as indicated in (**b**). The intensity I<sub>P</sub> in point P on the surface is inversely proportional to the distance R<sub>1</sub> between S<sub>1</sub> and P and to the distance R<sub>2</sub> between S<sub>2</sub> and P, i.e., the voltage in P is V<sub>P</sub> =  $kS_1/R_1 + kS_2/R_2$ , where k is a proportionality constant that accounts for the medium properties.

#### 2.3. Teaching Differential/Bipolar Detection, Innervation Zones, and the End-of-Fiber Effect

The electrical image describing the instantaneous distribution of the sEMG voltage is called "monopolar" when the voltage of each point is referred to a remote reference taken as zero (i.e., the resting water level in Figure 4). This image is "sampled" in a number of points in space (electrodes), whose voltages can be combined and manipulated to achieve specific purposes. Figure 7 shows a set of the simplest of these manipulations, called "spatial filters" (see Sections 2.3.1 and 2.3.2). The time-varying image generated by the firing of MU1 and MU2 is sampled in space, along a line, by a 16-electrode linear array and the differences  $V_A = V_1 - V_2$ ,  $V_B = V_2 - V_3$ , ... up to  $V_Q = V_{15} - V_{16}$  are presented as output signals. Note that these differences are calculated by the 15 differential amplifiers that compute  $V_{+} - V_{-}$ . If the linear electrode array were inverted (with Electrode 16 at the top and Electrode 1 at the bottom), the polarity of all signals would be inverted. Note that the two MUs have different IZs, located below Electrodes 7-8 (Channel G) and Electrodes 8-9 (Channel H), and their fibers terminate under the two ends of the linear electrode array. Therefore, it is possible to estimate the MU fiber length, as well as the CVs and the IZs. Note, for MU1, the lack of propagation in the action potentials that are detected by Amplifier A (compared to B) and Amplifier Q (compared to P). This lack of propagation indicates the end of the fibers (better defined later in this subsection) and, therefore, their length spans the distance between Amplifiers B and P. Additionally, note that the channels on one side of the IZ have a polarity while those on the other side have the opposite polarity. The amplitudes of the single differential (SD) signals detected above the IZs are near zero.



**Figure 7.** Schematic drawing of the propagation of two motor unit action potentials (MUAPs) under a linear electrode array that is aligned with the fiber direction. MU1 discharges at time  $t_1$  and MU2 discharges at time  $t_2$ : example of the estimation of the innervation zones and the calculation of the muscle fiber conduction velocity (CV). (See animation in Supplementary Materials, Sup4). Reproduced from (https://www.robertomerletti.it/en/emg/material/teaching/module6).

Note that signals from double (or multiple) IED values may be obtained either by taking the difference between the corresponding monopolar signals or by adding adjacent SD signals, since  $V_1 - V_3 = (V_1 - V_2) + (V_2 - V_3)$ . Double differential (DD) signals can be obtained from the difference between adjacent SD signals.

These are useful properties of differential detection and allow for the easy identification of the IZ of an MU as the region where the action potentials change polarity. This information can have clinical applications (see Section 4.4).

Figures 8 and 9 depict experimental differential recordings from two biceps brachii, both obtained with linear electrode arrays. Nine MUAPs can be identified in Figure 8 within a 100-ms time window. It is evident that all of the detected MUs are innervated under Channel 8 and have about the same length. Minor differences between their conduction velocities cannot be detected visually and require calculations using software. Figure 9 shows three MUAPs generated by different MUs within a 45-ms time window. Two MUs are innervated between Channels 8 and 9 and one is innervated under Channel 4. Note that (a) the MU fibers are longer than the array and (b) the MUAP propagation is monodirectional upward from Channel 9 to 11 and downward from Channel 3 to 1 and bidirectional between Channels 8 and 4 (downward for MU1 and MU2, upward for MU3). The average value of the muscle fiber CV (global CV) cannot be estimated using Channels 4 to 8. The decomposition of the sEMG into the constituent MUAP trains is necessary to estimate the CVs of individual MUs [23].

When the action potential reaches the end of a muscle fiber, it stops generating a transient signal; this is referred to as the end-of-fiber (EOF) effect, which is described in detail in https://www.robertomerletti.it/en/emg/material/teaching/module5 and https://www.robertomerletti.it/en/emg/material/teaching/module9. This phenomenon is modeled by describing the transmembrane current distribution with a tripole that is made of two opposing dipoles (d<sub>1</sub> and d<sub>2</sub>), as indicated in Figure 10a. When the tripole reaches the end of the fiber, the first dipole shrinks (Figure 10b, times t<sub>1</sub> and t<sub>2</sub>) and disappears (Figure 10b, time t<sub>3</sub>), leaving the second dipole d<sub>2</sub> whose field is stronger because it is no longer partially canceled by d<sub>1</sub>. In turn, d<sub>2</sub> shrinks (Figure 10b, time t<sub>4</sub>) and disappears (Figure 10b, time t<sub>5</sub>). This model explains the presence of non-traveling waves in the monopolar signals, as indicated in Figure 10c, but the exact mechanism is not known. Non-propagating signal components are often observed in monopolar sEMG signals and are



reduced by bipolar (SD) detection, as well as other "common mode" signals. An example of the EOF effect is provided in the simulation video of Figure 11.

Figure 8. Example of experimental sEMG signals recorded with 15 single differential channels from a biceps brachii. We can see the "signatures" of at least 9 MUs, locate their innervation zones, and estimate their length and conduction velocity (CV). The slopes of the dashed lines are the CVs of the motor unit action potentials (MUAPs). This piece of information cannot be obtained by one pair of electrodes and requires detection by an array of electrodes placed along the muscle fiber direction, as in Figure 7. IED = interelectrode distance. Modified from https: //www.robertomerletti.it/en/emg/material/teaching/module6.

50

time (ms)

100



Figure 9. Example of experimental sEMG signals recorded with a 12-electrode array providing 11 single differential (SD) channels from a biceps brachii. In the 45-ms window, we can see the "signatures" of three MUs and locate their innervation zones. In this case, global conduction velocity (CV) can only be estimated in the proximal and distal regions of the muscle. In the middle region (Channels 4 to 8), some motor unit action potentials (MUAPs) propagate upward and some downward. The concept of global, or mean, muscle fiber CV is meaningless in this region and can only be estimated for individual MUs after extracting their MUAPs by means of sEMG decomposition [23]. IED = interelectrode distance. Reproduced from https://www.robertomerletti.it/en/emg/material/teaching/module6.

#### 2.3.1. Teaching Action Potentials through Modeling

Despite their many limitations, models of sEMG are powerful teaching tools because they can show, with some approximation, the effects that changing individual fiber (or MU) parameters have on the sEMG signal [24,25,27,28]. Additional examples of this use of models for teaching muscle electrophysiology and interpreting sEMG features are provided in https://www.robertomerletti.it/en/emg/material/teaching/module9.



The electric field generated by the dipole at  $t_3$  is stronger than that generated by the tripole at  $t_1$  and  $t_2$  because an opposing dipole disappeared and only one remains.

**Figure 10.** (a) Schematic representation of the transmembrane current as two opposite dipoles (a tripole) of current, which implies the representation of the action potential as a triangular waveform; (b) extinction of the tripole at the end of the fiber, where the dipole  $d_1$  shrinks (at times  $t_1$  and  $t_2$ ) and disappears (at time  $t_3$ ), leaving dipole  $d_2$  whose field is no longer partially canceled by  $d_1$ . As  $d_2$  shrinks and disappears (at times  $t_4$  and  $t_5$ , respectively), the surface potential disappears; (c) example of a real monopolar motor unit action potential (MUAP) whose propagation phase is followed by a non-propagating transient due to the end-of-fiber (EOF) effect. This transient is partially removed by bipolar (SD) detection. In general, the sEMG signal includes propagating and non-propagating components, whose presence requires caution in selecting the signal segment to use for the estimation of muscle fiber CV. IED, interelectrode distance. Modified from https: //www.robertomerletti.it/en/emg/material/teaching/module5.

stronger EOF effects.

Figures 12–15 show the simulation of an action potential propagating along a single fiber that is innervated in the middle and detected by a 16-point-like-electrode array with an IED of 5 mm. The monopolar and SD signals are presented. The effects of fiber depth, lateral displacement with respect to the electrode array, the misalignment angle between the fiber and the electrode array, and conduction velocity are depicted. The effects are about the same for an MUAP when the IZ of the MU is narrow.

According to the model, the decrement of sEMG amplitude (root mean square, RMS) with increasing depth within the muscle is similar to that due to the increasing thickness of subcutaneous tissue. Experimental findings from obese subjects have indicated that sEMG is still detectable with subcutaneous tissue thicknesses of 20–30 mm [30]. The lowering of the electrode–skin interface noise may increase detectability at greater distances in the future [31].

Simulations of pathological fibers and MU alterations are lacking in the literature and provide an interesting research field for clinical application, the improvement of the available models, and application in the teaching and clinical interpretation of sEMG signals. A review of the use of sEMG in neuromuscular pathologies is provided by Hogrel [32].

#### 2.3.2. Spatial Filters

The only potential map that is generated on the skin by a biological signal is the monopolar map. Other maps may be obtained using analogue or numerical combinations of monopolar electrode potentials, as indicated in Figure 16. These combinations, or montages, are usually "differential" and have different properties, with the main difference being the reduction in the detection volume, i.e., greater spatial selectivity. However,



"low-pass spatial filters" may be obtained using larger electrodes or averaging blocks of monopolar signals to simulate larger electrodes.

**Figure 11.** Slow-motion simulation of the generation, propagation, and extinction of an action potential produced by a single fiber whose neuromuscular junction is not in the middle of the fiber. Observe the two end-of-fiber effects taking place at different times at the fiber-tendon junctions T1 and T2. They generate two sharp non-propagating transients in the sEMG signal. For a motor unit, the end-of fiber effect(s) may be smaller and less sharp, depending on the scatter of the fiber terminations. (See animation in Supplementary Materials, Sup5). Reproduced from https://www.robertomerletti.it/en/emg/material/teaching/module9.



**Figure 12.** Effects of fiber depth within the muscle: skin thickness = 1 mm, subcutaneous (fat) tissue thickness = 3 mm, interelectrode distance = 5 mm. Observe how, in the monopolar recording, the end-of-fiber effect decreases slowly and becomes greater than the propagating component for depths near 8 mm within the muscle. This contributes to the crosstalk signal that is present on nearby muscles. (See animation in Supplementary Materials, Sup6). Reproduced from https://www.robertomerletti. it/en/emg/material/teaching/module9.



**Figure 13.** Effects of lateral displacement (transversal distance) between the electrode array and the fiber: same conditions as in Figure 12. Observe that the end-of-fiber effect is visible for distances of greater than 10–15 mm, where it becomes larger than the propagating component. This contributes to crosstalk between muscles that is still detectable at lateral distances of 20–30 mm from the edge of a muscle, mostly as a non-propagating component. (See animation in Supplementary Materials, Sup7). Reproduced from https://www.robertomerletti.it/en/emg/material/teaching/module9.



**Figure 14.** Effects of the rotation of the fiber with respect to the array (or vice versa): the rotation center is the center of the fiber that coincides with that of the array and with the innervation zone (Channel 8). Observe how the detected propagating action potential decreases as it moves along the fiber. This is because the source moves away from the electrodes. The case of a fiber inclined and displaced laterally with respect to the electrode array (not presented) is frequent for muscles that have a fan-like or pinnate structure. More information is provided in [29]. (See animation in Supplementary Materials, Sup8). Reproduced from https://www.robertomerletti.it/en/emg/material/teaching/module9.



**Figure 15.** Effects of changing muscle fiber conduction velocity (CV). Observe that the width of the action potential and the end-of-fiber effect become narrower and the slope of the pattern increases as the CV increases. The observed changes affect both the single sEMG channels and the array of signals. When a single channel is recorded, the effects are not detectable. (See animation in Supplementary Materials, Sup9). Reproduced from https://www.robertomerletti.it/en/emg/material/teaching/module9.



R

R

**Figure 16.** Electrode configurations or "montages": (**a**) monopolar (any combination or spatial filter can be obtained by a grid of monopolar signals); (**b**) bipolar or single differential (SD); (**c**) array of SD signals; (**d**) double differential (DD); (**e**) normal (perpendicular) DD (NDD) or Laplacian configuration. Other configurations are possible. The reference electrode R must be on a region with no sEMG activity and away from the muscle of interest. It should not be on the same muscle. The NDD system is widely used in EEG because of its selectivity. Some recently developed systems do not need a reference electrode. A is the amplification or gain of the differential amplifier. Modified from https://www.robertomerletti.it/en/emg/material/teaching/module6.

#### 2.4. Two-Dimensional (2D) sEMG Representation and sEMG Images

With respect to linear electrode arrays, bi-dimensional electrode arrays or grids provide additional information in the muscle fiber transversal direction and allows for the better discrimination between MU "signatures" and orientation, thereby facilitating the decomposition of the sEMG into the constituent MUAPs, as well as the localization of the active MUs in space and the size of the MU territory (i.e., the spatial distribution of the muscle fibers belonging to an MU).

Figure 17a shows a 2D grid of  $8 \times 4$  electrodes. Two of these grids are applied to the right erector spinae and two are applied to the left, as in Figure 17b,c, which shows the values (see the color scale) of the RMS of the differential voltage that is detected by each longitudinal pair of electrodes of the two grids placed from proximal to distal, on the right side, over a time window of 1 s. Figure 17d shows the interpolated version of the map and the direction of the fibers. The orientation of the fibers is outlined. It is also evident that the quality (and correctness) of the interpolated map is strongly dependent on the IED. In the case of sEMG images, an IED of 10 mm is a borderline value [33] and provides an image reconstruction that has minor inaccuracies (see Section 4).



Electrode diameter = 3 mm

**Figure 17.** Example of sEMG distribution on the erector spinae of a healthy subject: (a) grid of  $8 \times 4$  electrodes (electrode diameter = 3 mm; center-to-center interelectrode distance IED = 10 mm). (b) two such grids are applied on each side of the spine to produce an array of  $15 \times 4$  single differential signals whose root mean square (RMS) value (over a 1s epoch) is depicted in (c), according to the color scale; (d) interpolated image, obtained from (c), of the RMS distribution with an indication of the orientation of the fibers. The blue area in the middle corresponds to the innervation zone of the superficial motor units.

The concepts presented in this section raise a number of practical questions, such as "What should the size (i.e., the electrical contact area) of the electrodes and the IED be to allow for the correct reconstruction of a sampled image by interpolation?", "Are there optimal values for the size and IED for maximal information extraction?", "How big should an electrode grid be?", "Where should it be located?", "What are the advantages and disadvantages of monopolar versus SD images?", and "When is a single electrode pair preferable to HDsEMG?". These and many other issues are addressed in Sections 3 and 4 and in sEMG textbooks and articles [22,33,34] (see https://www.robertomerletti.it/en/emg/material/books/ for a more complete list), which provide a comprehensive mathematical analysis of these problems. Sections 3 and 4 attempt to answer these questions in a simple way.

#### 3. The Role of Electrode Size and Interelectrode Distance in Bipolar sEMG

Electrodes that are applied to the skin are not point-like, as is often assumed in modeling. From the theoretical signal processing point of view, the full information that is contained in the superficial distribution of EMG potential can only be obtained when proper sampling in space is performed (and when spatial aliasing and smoothing are avoided, see [33] for more details). This implies the need for small electrode sizes and small IEDs (a) to avoid the substantial filtering that is introduced by the transfer function of the electrode pair and (b) to reduce crosstalk. In turn, this implies the need for the selective detection of the most superficial MUs.

However, in many clinical applications, only a portion of this information is useful and greater detection volumes may be desirable, which provide larger signals at the price of lower selectivity and possible crosstalk from nearby muscles [35,36]. This also applies to single electrode pairs, in which small EEG electrodes (e.g., http://www.gereonics.com/eegelectrodes.html may be preferable to the larger pediatric ECG electrodes that are often used. Commercially available active sEMG sensors that carry two parallel silver bars, 10 mm long and 10 mm apart, also provide a good solution (e.g., http://delsys.com/sensors.

A pair of large electrodes (diameter = 10 mm) and a large IED (e.g., 20 mm) introduce the attenuation of the highest harmonics in the signal (see Figures 2.5 and 3.11 of [35]), as explained in the following subsections. For most clinical applications of bipolar sEMG, this low-pass filtering is acceptable. Conversely, the reduced selectivity, i.e., the risk of crosstalk, may be critical and lead to misinterpretations. Thus, miniaturized electrodes should be used when assessing muscles with small cross-sectional areas, such as forearm muscles and children's muscles.

In sport applications, sEMG can be used to compute the so-called localized myoelectric manifestations of muscle fatigue. Since this indicator is based on the spectral features of the signals, the filtering effect of the detection system (i.e., electrode size and IED) cannot be ignored. Users must be aware that their results depend on the electrode configuration and cannot be compared to those obtained using different electrode geometries, e.g., in the literature. The user must also be aware that the electrode size is the electrical contact area (possibly mediated by gel) and does not include the surface of the external adhesive disks that keep the electrode in place when present.

#### 3.1. Electrode Size and Its Filtering Effect

An electrode has a highly conductive surface, which forces all of the points below it to have the same voltage. This voltage is approximately the mean value of the voltages at the points below the electrode surface (without the electrode). This averaging effect is a form of low-pass filtering, which reduces the "sharpness", or high-frequency content, of the detected voltage. To understand the relevance of this effect using a simple example, consider the animation in Figure 18a, in which a signal propagates under an electrode of length L, which provides the average value of the signal over L in the direction of propagation versus time (Figure 18b). The electrode attenuates and smooths the signal, thereby introducing low-pass filtering. Observe the cancelation of the periodic signal, which has a period in space (wavelength) that is equal to L/2. Figure 19 shows this effect on a real monopolar sEMG signal. In practice, since a larger electrode increases the detection volume, the amplitude may not be reduced (it may actually increase because a larger number of MUs is covered by the electrode) but the spectral alteration (smoothing) remains.

As indicated in the literature [33], sEMG electrodes with diameters of up to 3 mm are acceptable, while diameters greater than 5 mm modify the shape and the spectral features of the MUAPs of the most superficial Mus, thereby making these MUAPs more similar to those of deeper MUs.

Again, it is clear that electrodes that have diameters of 5 mm are barely acceptable, while electrodes that have diameters of  $\geq 10$  mm introduce substantial filtering and modify the waveshape, amplitude, and spectral features of the MUAPs and, with that, the interferential sEMG signal, particularly in lean subjects with many MUs that are located close to the

skin surface (see Section 3.2). Nevertheless, these alterations may be acceptable or irrelevant for a number of applications whose objectives are little affected by these factors, such as the analysis of gait or studies on coordination with one, properly positioned, electrode pair per muscle.



**Figure 18.** Effects of electrode length on a signal: (**a**) the green signal propagates in space under the gray fixed electrode of length L, generating the instantaneous signal indicated by the red bar. The signal evolves in time, as depicted in panel (**b**). (See animation in Supplementary Materials, Sup10). Courtesy of Dr. Alberto Botter.



**Figure 19.** Electrode filtering of a monopolar experimental sEMG signal. (a) moving average performed by a round electrode over a propagating monodimensional signal, (b) a signal collected using a pin electrode is filtered by the transfer function of electrodes of different sizes (in one dimension only), producing a number of outputs that correspond to different electrode diameters. The signal is low-pass filtered and attenuated because only the filtering effect of the different electrode sizes are accounted for. In practice, a larger electrode covers more fibers, so the amplitude may not be reduced (it may actually increase) but the spectral alteration due to filtering remains and alters the amplitude and spectral features of the signal. Modified from https://www.robertomerletti.it/en/emg/material/teaching/module9.

#### 3.2. Interelectrode Distance and Its Filtering Effect

The ubiquitous differential detection (SD) provides the voltage difference (versus time) between two electrodes that are spaced by an IED in the fiber direction. Figure 20 demonstrates how the IED affects the detected SD waveform and, consequently, the sEMG

features. A simple triangular waveform propagating under the two electrodes systems A–B, spaced by an IED of  $e_1$ , and C–D, spaced by an IED of  $e_2$ , is provided as a teaching example. The results in time are very different. If  $e_2$  were equal to the wavelength of the propagating signal, the output in time would be zero at all times, thereby generating an artificial "dip" in the amplitude and power spectrum of the time signal for that particular frequency. For this reason, the IED must be smaller than half of the shortest wavelength that contributes to the sEMG signal, which is about 10 mm (corresponding to 100 cycles/m) for MUs that are close to the skin surface [33]. According to mathematical theory (Nyquist sampling theorem, not discussed here), the spacing between two electrodes should therefore be less than 5 mm. However, due to the spatial low-pass filtering effect of the volume conductor, the minimum wavelength that contributes to the potential distribution is muscle- and subject-dependent. Deeper muscles show skin potential distributions with longer wavelengths. The potential distributions of people with a higher body mass index have longer wavelengths than those of people with a lower body mass index. Therefore, in practice, values of up to 8–10 mm may be acceptable, with minor signal alterations because of the small and long "tails" of the sEMG power spectrum. The effects of electrode size could be added to the simulation in Figure 20 using wider slots and taking the average value of the signal within each slot. See also Figure 18.



**Figure 20.** Concept of spatial filtering that is introduced by bipolar (SD) detection. Panels (**a**,**b**) show how the same triangular voltage profile propagating under an electrode pair can produce very different SD signals, depending on the interelectrode distance (IED). The same happens for the sEMG. To avoid this effect in sEMG detection, IEDs should be  $\leq 5$  mm (but values of up to 8–10 mm introduce signal alterations that are acceptable for most applications). This is particularly important for detection using electrode grids. It is less important for clinical bipolar detection, as long as the same electrode size, IED, and location are used in subsequent measurements for comparing sEMG features, for example, in pre- and post-treatment or intervention. (See animation in Supplementary Materials, Sup11). Reproduced from https://www.robertomerletti.it/en/emg/material/teaching/module9.

In clinical practice, SD (bipolar) sEMG detection is often performed using electrodes that have diameters of 10 mm or more and an IED of 20 mm or more. The combined effect of these values has been discussed in a tutorial and a textbook (see Figures 13 and 14 of [19,35]). Of course, a signal may be obtained using any combination of electrode size and IED and two signals that are detected by different detection systems may look visually alike but have different spectral features because of the electrode low-pass filtering effect and the "dip" effect, which strongly modify the signal spectrum [19,33]. Large IED values (IED > 10 mm) are often preferred because they detect sources from a larger volume and provide larger signals. However, they are more sensitive (a) to crosstalk from neighboring muscles and (b) to the effect of MUAP generation at the IZ and MUAP extinction at the muscle–tendon junction (EOF effect). In addition, for the reasons presented above, large IED values (IED > 10 mm) are not suitable for estimating muscle fiber CV, for which at

least four electrodes on the same side of the IZ are usually required as well as the most suitable DD derivation. Finally, the features of different sEMG signals cannot be compared when the signals are collected with electrodes of different sizes, distances, and locations with respect to the IZ(s). The effects of these factors on the amplitude and spectral sEMG features of the upper trapezius and knee extensors are clearly explained in [37,38].

The above considerations apply in general and also specifically in the case of muscles that are located very close to the skin surface with thin skin and subcutaneous tissue. In the case of more distantly located muscles or subjects with a higher body mass index, these considerations may be somewhat relaxed because of the low-pass filtering effect introduced by the increased distance between the source and detection points. This factor reduces the bandwidth of the sEMG signal in space and time. This issue has been investigated by Minetto et al. [30], who concluded that because of the relevance of non-propagating components (due to the EOF effect and crosstalk), muscle fiber CV (when of interest) cannot be reliably estimated in these cases: "Interelectrode distance greater than 10 mm is not only recommended but required in sEMG detection from the vastus medialis and lateralis (VM and VL) of obese subjects to counteract the sEMG amplitude attenuation due to the increased thickness of the subcutaneous layer." Since, in these cases, the sEMG power spectrum is narrower and the shortest wavelength in space is longer, it can then be concluded that IEDs of 5–10 mm should be used to investigate the superficial muscles of lean subjects while IEDs of 10–20 mm could be used in studies concerning muscles that are located more distantly, especially in subjects with a greater body mass index.

Similar considerations apply to the issue of electrode size. The option to select IED implies adjustable selectivity, which is not possible with coaxial needles. The issue of electrode location, size, and IED becomes particularly important when electrode grids are used, spatial interpolation is applied or individual MUAPs are to be analyzed.

#### 4. Factors That Can Affect sEMG and Its Physiological Interpretation

The distribution of sEMG potential is a time-evolving monopolar image (a movie) whose full information content can only be recovered and used when the image is properly sampled in space and time. However, most current clinical applications are based on traditional bipolar (SD) readings since the obtained information is, in general, appropriate for answering clinical questions with an adequate compromise between the in-field feasibility of the activity detection of many muscles and the reduction in information on the active MUs.

It is important to understand that a bipolar recording is nothing more than taking two samples in space out of the monopolar map and plotting their instantaneous voltage difference versus time. When using bipolar detection, some of the strict requirements concerning the electrode size and IED of an electrode grid can be somewhat relaxed in favor of a larger detection volume and lower noise, at the price of lower selectivity and a greater risk of crosstalk. Electrodes that are located improperly or have large sizes and distances cause waveform changes (Figures 18–20) and an often-undetected sensitivity to crosstalk, as shown for the brachioradialis muscle by Merlo et al. in Special Issue [39].

In addition, during dynamic contractions, the muscle moves under the electrodes and this may cause changes in the sEMG amplitude that are only due to the changes in the relative IZ position. These issues are rarely discussed in the clinical literature or addressed in schools or courses and sEMG users seem not to be aware of them. Consequently, incorrect conclusions may be reached by unaware users.

#### 4.1. Skin Treatment

Proper skin treatment is important for both bipolar and HDsEMG recordings. In HDsEMG, the relatively large number of electrodes increases the likelihood of bad contact and requires the accurate treatment of the skin (e.g., shaving, rubbing with peeling cream, and then cleaning with a damp cloth), preferably without using alcohol, which leaves skin dry [31]. This requires some training in skin treatment and proper grid application, with

uniform pressure on each electrode and no wrinkles. When conductive paste or gel is applied to arrays or pairs of electrodes, gel bridges between the electrodes must be avoided. The performance and repeatability of results rely on proper electrode choice, placement, and management, keeping in mind that small electrodes are noisier and more selective than large electrodes but introduce less signal filtering, as discussed above and indicated in Figures 18 and 19. It is also important to point out that the two electrode–skin impedances of an electrode pair must be small and similar to reduce power line interference [20].

#### 4.2. Movement of the Muscle under the Electrodes

Figure 21 addresses the important problem of muscles sliding under the skin during dynamic contractions. The example shows two interpolated images obtained using an electrode grid that is placed on a biceps brachii. When the forearm flexes from  $150^{\circ}$  to  $70^{\circ}$ (internal angle between arm and forearm), the biceps muscle shortens under the array and the EMG amplitude map changes from Map 1 to Map 2. Considering one operator placing the white pair of electrodes and another operator placing the black pair, the first would conclude that elbow flexion increases the sEMG amplitude but the second would conclude the opposite. The change in sEMG amplitude that is detected by the two electrode pairs when the elbow flexes is due to the change in geometry, not the change in muscle electrical activity. This is an important issue to consider during dynamic contractions and is likely to be one of the reasons for the lack of uniform results reported in the literature, in addition to electrode size and IED. Users should be aware of this phenomenon, which takes place in long fusiform muscle-spanning joints with large angle variations. Although only examples from biceps brachii have been reported, other muscles may show the same phenomenon. An atlas for proper electrode placement should always be used when dealing with bipolar sEMG. It should also be verified that the electrode pair remains on the same side of the muscle IZ(s) during the entire range of movement. Since this is impossible when using bipolar electrodes, atlases that indicate the regions where electrodes should not be placed have been made available [22].

#### 4.3. Identification of Innervation Zones

The identification of the IZ by means of linear electrode arrays or HDsEMG, as reported in Section 2.3, has more potential clinical applications other than being useful for supporting the proper positioning (or selection) of a pair of electrodes [40] (far from IZs and on minimum crosstalk areas), including the targeted injection of the botulinum neurotoxin, which is commonly used to treat disorders such as cerebral palsy, spasticity, and cervical dystonia [4,5,41].

When the toxin is applied to the IZ, a lower dose can be highly effective [4,5,42]. The use of HDsEMG to target the IZ could reduce the amount of injected toxin and, consequently, limit the treatment cost [43]. An atlas of muscle innervation zones is available for most superficial muscles [22].

The identification of the IZ is also a useful tool for planning surgery, for example, in the case of episiotomy during child delivery. The innervation to the external anal sphincter muscle may be damaged due to episiotomy, thereby increasing the chances of future fecal incontinence [44]. The European project OASIS and the multicenter study TASI exploited multichannel sEMG probes to identify changes in the external anal sphincter innervation in women who underwent episiotomy during labor. It was found that (a) the distribution of IZs is highly variable across individuals [45] but can be identified in a repeatable way across days [46] and (b) the number of IZs significantly decreases in the quadrant where the episiotomy was performed [45]. Therefore, the identification of the IZs prior to delivery could be a preventive measure to guide the episiotomy incision (when necessary) and to suggest the ways or sides with the least risk. Figure 22 depicts the probe that was used, one significant result, and the statistical results from 82 cases of episiotomy.

96 mm



sEMG differential RMS maps

**Figure 21.** Example of a biceps brachii muscle shortening under the skin during a dynamic contraction. When the forearm flexes from 150° to 70° (internal angle with respect to the arm), the biceps muscle shortens under the array and the map changes from Map 1 to Map 2. Considering the white and the black pair of electrodes as being positioned by operators A and B, the white pair would indicate that elbow flexion causes an increase in the sEMG amplitude whereas the black pair would indicate the opposite. Great caution must be used during the placement of single electrode pairs for the study of dynamic contractions because the muscle geometry changes taking place under the electrodes cause sEMG variations that may be erroneously attributed to changes in the neural muscle drive when electrodes are placed on or near to the innervation zones (IZs). Adapted with permission from Figure 6.5 of Ref. [22]. Copyright Springer Verlag Italia, 2012.



**Figure 22.** A study of the anal sphincter pre- and post-episiotomy: (**a**) the anal probe with 16 silver electrodes equi-spaced along the circumference; the detection of the innervation zones (IZs) in the same woman before (**b**) and after (**c**) delivery with episiotomy. The black cylinders represent the neuromuscular junctions and the gray arcs depict the identified motor units (MUs). In this subject, no MUs were detected on the side where the episiotomy was performed. The results from 82 cases of episiotomy are shown in (**d**). Fewer MUs were detected on the right ventral (RV) quadrant, where the episiotomies were performed, suggesting damage to the innervation following surgery. Note: the information collected before the delivery was not used to plan the surgery. Quadrants: LV, left ventral; LD, left dorsal; RD, right dorsal; RV, right ventral. Adapted with permission from Figures 1, 2, and 4 from Ref. [45]. Copyright The International Urogynecological Association, 2014.

Despite the literature in this field [45–47], the clinical use of sEMG is still absent because of the lack of information and teaching for obstetricians, midwives, and pelvic floor therapists.

#### 4.4. Examples of HDsEMG Applications in Neurophysiology

A comprehensive overview of all of the clinically possible applications of single channel and HDsEMG in neurology exceeds the purpose of this paper. Therefore, only a few relevant examples are highlighted in the following section to underline the need for training with a view to future applications. For a more comprehensive analysis, refer to the systematic reviews in [7,9,32].

HDsEMG was already being used in the early 1990s to differentiate the pathologies of neuronal origin from those of muscular origin. Ramerkers et al. [48] were able to show that when using HDsEMG with the NDD detection system (Figure 16), the resulting signals that are derived from patients with muscular or neuronal disorders exhibit disease-specific features.

Figure 23a shows the NDD spatially filtered signal as it manifests in neuropathies or myopathies. While myopathies are mainly characterized by a change in the shape of the MUAPs, neuropathies show a decrease in the detected number of active MUs. Simulation studies have indicated that in the case of myopathies, the degeneration of muscle fibers is responsible for the changes in the signal, whereas a loss of Mus leads to the characteristic changes found in neuropathies [26]. These typical changes in the spatially filtered HD-sEMG signal that characterize muscular or neuronal pathologies can be described by seven parameters and enable correct diagnostic conclusions with an average accuracy of 97% [49]. This means that the possible diagnostic yield of the non-invasive HDsEMG technique, for superficial muscles, is in the range of needle electromyography [50,51]. Other applications refer to the MU number estimation, the quantification of fasciculations [52], HDsEMG changes in the muscle structure, and MU recruitment of stroke patients in pathophysiology studies [53].

As HDsEMG is a non-invasive method, it is particularly useful for monitoring the progress of neuromuscular disorders in children. Furthermore, for superficial muscles, HDsEMG provides additional information about the MUs and their recruitment, which is not accessible through needle EMG but could be relevant for diagnostic purposes. This includes information about the CVs of MUAPs along the muscle fibers, as well as the spatial distribution of excitation within a muscle.



**Figure 23.** Characteristic changes in the normal double differential spatially filtered (Figure 16) high-density sEMG (HDsEMG) signal associated with different disorders: (**a**) myopathies lead to a change in the shape of the motor unit action potentials (MUAPs) while neuropathies lead to a decrease in the number of MUAPs contributing to the signal; (**b**) neuromuscular disorders involving the upper motor neuron, such as stroke, are characterized in the spatially filtered HDsEMG signal not only by fewer active motor units, but also by a lower variability in firing rate [49,54]. For a better representation of these effects, (**a**,**b**) have different temporal scales.

Figure 24 indicates how the MUAP CV changes as children develop. The most significant age-related change in MUAP CV occurs at an age of up to 5 years. From an approximate age of 15 years, the MUAP CV remains constant. The CVs of the MUAPs are significantly reduced in patients with Duchenne muscular dystrophy, compared to healthy children [55], and in patients suffering from carpal tunnel syndrome [56].

Many other neurological applications of HDsEMG are clinically relevant. Among these are hypokalemic periodic paralysis, pathological fatigue, and MU size estimation, presented in the reviews by Zwarts et al. [7], Drost et al. [9], and Hogrel [32], along with a discussion of the advantages and limitations of HDsEMG.

Numerous applications of HDsEMG are being developed in research labs to decode the neural drive to muscles by means of the decomposition of the sEMG into the constituent MUAP trains. These methods are based on the early work of Holobar and Zazula [23] and are ready for clinical evaluation.

Finally, it is fair to say that medical devices can only be used within their "intended use" (what can and cannot be carried out with that device), which is indicated by the producer in the first pages of their user manual. Any diagnostic tool requires a class II medical device certification in Europe, an FDA approval in the USA or a similar certification in other countries. Users and researcher should verify the certification level before buying or using sEMG devices for specific applications.

#### 4.5. Issues of Bipolar sEMG Detection in Neurorehabilitation Applications

In neurorehabilitation, the use of motion analysis with sEMG clearly influences treatment planning in multiple patient populations and can lead to better outcomes when properly used [57]. Gait analysis is a functional evaluation that provides a quantification of the observed performance and identifies motor deficits and the underlying causes of the observed pathological patterns [58–60]. This evaluation allows for the formulation of a precise and targeted therapeutic plan, such as surgical, pharmacological, orthotics or physiotherapy plans [57,58,61–64]. In the study of muscle activation during the gait of neurological patients, it is of paramount importance to avoid common errors that could possibly lead to incorrect clinical decisions [39,65,66]. The literature provides guidance on how to properly perform an examination so that it can be reliable and used for functional assessments and clinical evaluations [67].



**Figure 24.** Change in the mean conduction velocities of motor unit action potentials (MUAP CVs) with age in healthy subjects and in patients suffering from Duchenne muscular dystrophy or spinal muscle atrophy. The solid black line is the regression curve for the healthy subjects. The gray area contains 95% of all healthy subjects. Adapted with permission from Ref. [55]. Copyright Muscle and Nerve, 1997.

The main errors in the use of sEMG for the evaluation of neurological patients are attributable to the type of electrodes, the placement protocol, and the presentation of the data that are chosen. The traditionally used SENIAM recommendations for electrode placement [68] may not consider the problem of crosstalk, as indicated in Figures 25 and 26. In order for the evaluation to be clinically usable, it is necessary for the chosen electrode placement protocol to guarantee the selectivity of the analyzed muscle activation, with small electrodes placed on the so-called minimum crosstalk areas [39,69–71] and on the same side as (and far away from) the IZs [22].



**Figure 25.** Envelopes of bipolar sEMG (one plot per stride for 10 strides) from nine electrode pairs obtained by placing a  $4 \times 3$  grid of electrodes,  $15 \times 18$  mm in size, over the anterior aspect of the leg. The medial column of electrodes is placed on the edge between the tibialis anterior muscle and the tibia bone. The gray bands indicate the presence of activity in the mid and terminal stances due to signals coming from the peroneus longus. Adapted with permission from Ref. [66]. Copyright Journ. of Electromyography and kinesiology, 2007.



**Figure 26.** Example of sEMG signals acquired during gait analysis of the leg of a healthy 10-year-old boy by means of pairs of circular electrodes: diameter of the conductive region = 10 mm; detection area = 78.5 mm<sup>2</sup>; center to center distance = 20 mm. Crosstalk on the tibialis anterior from the plantar flexors is outlined in purple. Reproduced from Ref. [72].

The width of the electrodes and the IED can also play a crucial role, especially when recording the activity of small muscles, such those as in children. The disposable ECG pre-gelled electrodes that are normally used in clinical evaluations are often too large. When the electrode contact area is wider than the target muscle, it can cover more than one muscle, thus reporting an incorrect activation time due to crosstalk.

Finally, it is important that the reports of the examinations carried out contain raw traces to distinguish the sEMG signal from other possible artifacts [73] that are very frequent in the study of neurological patients with movement alterations. In fact, the representation of the sEMG signal in the gait cycle through the envelope, although more synthetic, can be dangerous and misleading. It is important to always visually verify the quality of the raw sEMG signals since after the computation of the amplitude envelope, artifacts and interferences are no longer detectable and cannot be eliminated [74].

The issue of proper signal representation and scaling (of the electrodes and not the output of the amplifier) may seem obvious and trivial, but it is not [73]. Figure 27a shows a number of individual sEMG bipolar channels recorded from muscles in the left leg of a subject with a continuous activity of the rectus femoris (L\_RF). A more careful analysis reveals that the operator selected the "autoscale" option of the software instead of the "fixed scale" option. Figure 27b shows the same channels with the same scale (0.5 mV/div for the input signal), but the picture is very different. In addition, the "autoscale" option automatically selects the scale of the L\_PL muscle on the basis of the artifact outlined in yellow.

#### 4.6. Issues in Ergonomics, Occupational Medicine, and Sport Sciences Applications

Surface EMG is fundamental for the proper design of workstations and chairs, the planning of sport training, and the understanding and prevention of neck, shoulder, and back pain [15,75–77].

Ergonomics, occupational medicine, and sport sciences require specific competencies and training in sEMG. These areas are affected by the barriers and difficulties that have been discussed elsewhere [78]. The early NIOSH publication by Soderberg [79], the book by Kumar and Mital [80], the review by Marras [81], and the recent works of Gazzoni et al. [82] and Varecchia et al. [83], among many other contributions, underline the importance of teaching the proper use of sEMG in ergonomics and occupational medicine. Similarly, the early work of Clarys [84], Chapter 9 of Electrodiagnosis in New Frontiers of Clinical *Research* [85], and the recent work of Felici [77], among many other works, outline 30 years of efforts in the field of sEMG use in sports and exercise. Following this long history of research, "There is a critical need to ... instruct teachers across the disciplines of these fundamentals ... and give access of these developments to the current generation of students, which would improve the translation in order to potentially predict and cure pathologies of the neuromuscular function. This has an impact, for example, on the student awareness level of some very simple points related to the correct collection of sEMG data. Issues such as electrode location, IZ detection, skin preparation, movement artifacts and sweating are very often ignored." [77].

In addition, Martin and Acosta-Sojo [78] indicated that "The application of sEMG requires knowledge (e.g., EMG theory, signal processing, and neurophysiology), experience (e.g., electrode placement and signal morphology), and time (e.g., experimental design and procedure). These essential components may be deepened in doctoral studies but are not provided in clinical curricula such as physiotherapy and occupational therapy. Hence, broadening or amending the educational programs of physiotherapists, occupational therapists, movement scientists, and ergonomists (as future clinical users of sEMG) should prove useful."

Further considerations for sEMG applications in sports and ergonomics are relevant and discussed in the extensive literature [83,86] but exceed the purpose of this work. In all studies, the proper acquisition of sEMG is fundamental to reaching consistent conclusions.

The published literature [19,20,36,87] contains summary tables of all of the main factors that may affect correct signal estimation and the basic guidelines to follow for proper EMG recording.



**Figure 27.** Bipolar sEMG recordings from eight muscles of the left leg during the gait analysis of a subject: (**a**) "autoscale on"; (**b**) same scale for all channels. A superficial visual analysis of (**a**) would lead to incorrect conclusions.

#### 5. Discussion and Limitations of the Work

As indicated in the introduction, this work does not present innovative research results concerning sEMG hardware, processing techniques or applications. Rather, it presents (a) a number of the "first level" fundamental concepts of sEMG detection and interpretation that clinical operators should be familiar with and (b) some model-based teaching tools that could be useful to teach these concepts. All of these concepts can be found in textbooks and tutorials, in which they are presented within the mathematical framework that engineers are familiar with. This framework is unacceptable to most clinical operators, who have neither the need nor the background required to understand it. This work provides material and suggestions concerning the teaching of these concepts and their transfer into the educational curricula of clinical figures.

The same applies to other "second level" technical concepts, such as the spectral analysis of sEMG, the myoelectric manifestations of muscle fatigue, image segmentation and the definition of regions of muscle activity, the decomposition of interferential EMG into the constituent MUAP trains, the association between sEMG and biomechanical variables, sEMG in dynamic conditions, etc. These concepts are not discussed in this work, which is focused mostly on the fundamental issues of sEMG detection and simple model-based interpretation and teaching.

Surface EMG is very rich in information that can be extracted to draw conclusions about the coordination between muscles and muscle groups [88] and the anatomy, physiology, and pathology of a muscle and to monitor muscle changes following pathologies, training or therapeutic interventions. The proper understanding of sEMG generation and detection processes, as well as the visual understanding of the raw signals and the verification of good quality, is a fundamental preliminary step for the subsequent computer-assisted extraction of features (e.g., muscle fiber CV, amplitude and spectral variables, etc.) of sEMG to avoid the GIGO (garbage in, garbage out) effect.

A number of textbooks, recommendations, tutorials, consensus papers, and editorial projects have been published concerning these issues [17,19,20,22,34–36,89,90]. They include indications of the procedures to follow in order to (a) allow other researchers to reproduce the published work, (b) indicate whether or not results may be compared to those of others, and (c) promote the preparation of standards that account for clinical experiences. Despite the extensive literature concerning these recommendations, the published articles discuss a great variety of sEMG electrode sizes, distances, and locations. Sometimes, this information is not even reported or incorrectly reported, as indicated in the recent review by Merlo et al. in Special Issue [39]. A greater awareness of the problem is required by reviewers, journals, and schools, as well as by scientific and professional societies within the field.

A fundamental issue is the technology transfer from research labs to professional and medical schools, as well as to healthcare institutions. The simplification of concepts (as is often demanded by clinical operators and attempted here) and the availability of free teaching materials and recommendations are by far insufficient. As Rogers [91] notes, innovations are initially perceived as uncertain and even potentially negative. To overcome this uncertainty, most people seek out colleagues who have already adopted the new idea. Thus, the diffusion process is initiated by a few individuals who see the advantages in adopting an innovation and then spread the word among their colleagues or acquaintances, which is a process that takes years, regardless of the availability of textbooks, recommendations, publications, and teaching materials (which is often disregarded) [92]. Every change requires intellectual and learning efforts by the user and the view of such effort as a social, as well as economical, process [86,88,93] (see the EMG Reporting Standards of the International Society of Electrophysiology and Kinesiology, available online: https://isek.org/emg-standards, https://www.tudelft.nl/en/education/programmes/ bachelors/kt/bachelor-of-clinical-technology/, and https://www.tudelft.nl/onderwijs/ opleidingen/masters/technical-medicine/msc-technical-medicine/).

As in the case of other bioelectric signals, it is up to the clinical operator to promote and use newly available technical information that is suitable for properly answering clinical questions. Learning about this in school is important; however, the fields of rehabilitation technology and neuromuscular instrumentation are becoming so extensive that it is hard to fit even the basic concepts discussed in this work into the few academic years that lead to a degree in physiotherapy, sport and movement sciences or ergonomics. To some extent, this is true in other healthcare fields as well and has justified the training of new interdisciplinary professional health figures, such as the "clinical technologist" that trains at the Technical University of Delft (NL) (https://www.tudelft.nl/en/education/programmes/bachelors/ kt/bachelor-of-clinical-technology/, and https://www.tudelft.nl/onderwijs/opleidingen/ masters/technical-medicine/msc-technical-medicine/) within the BS and MS curricula for courses on "advanced signal acquisition", "sensing neurophysiological signals", and the "electrostimulation of the neurophysiological system". These new figures may solve the problem of the lack of time to teach measurement techniques, such as sEMG, to strictly clinical operators. This issue of the lack of time has been pointed out repeatedly in a number of surveys [17,94]. These new interdisciplinary figures may help to address problems that can be solved by technology that are not yet perceived as important by clinicians.

#### 6. Conclusions

Surface EMG technology is undergoing very fast development, which allows for the non-invasive investigation of movement, muscle function, and nervous system control strategies. This progress is much faster than the process of knowledge transfer, meaning that the knowledge gap is widening [3,32,92,95,96]. The correction of this situation requires the broadening of educational programs and the competence of the clinical users of sEMG, such as physiotherapists, occupational therapists, movement scientists, ergonomists, physical medicine and rehabilitation physicians, and clinical neurologists. This work indicates some options, tutorials, and tools that are available for motivating and training teachers, students, and clinicians for the challenges that they may face within the sEMG field. Existing tools and tutorials are mostly open-access and others are in preparation.

Many more materials should be developed, based on the experience of teachers and clinicians in the field, to provide explanations and descriptions of sEMG features (not discussed in this work) using analogies and user-friendly models.

The success of any future hardware/software technology to extract information from sEMG fundamentally depends on the ability of the operator to collect "good" sEMG signals from the desired muscles that are free from artifacts and power line interference and have minimal crosstalk from other muscles. The proper training and academic education

of operators is a necessary condition for the success of the available, as well as future, technologies within this field.

**Supplementary Materials:** The following supporting information can be downloaded at: https: //www.mdpi.com/article/10.3390/s22114150/s1, Figure 2: Sup1, Figure 4: Sup2, Figure 5: Sup3, Figure 7: Sup4, Figure 11: Sup5, Figure 12: Sup6, Figure 13: Sup7, Figure 14: Sup8, Figure 15: Sup9, Figure 18: Sup10, Figure 20: Sup11.

**Author Contributions:** I.C. and A.M.: conceptualization and contribution to writing Sections 4–6; C.D.-K.: contribution to writing Section 4; L.M.: preparation of Supplementary Materials and Power Point animations; S.M.: contribution to writing Section 4; R.M.: conceptualization and contribution to writing Sections 1–3; All authors: general revision and approval of the final version. All authors have read and agreed to the published version of the manuscript.

**Funding:** S.M. was supported by the European Union's Horizon 2020 research and innovation program under the Marie Sklodowska-Curie grant agreement number 846679 (INFANTPATTERNS).

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: Not applicable.

Acknowledgments: The authors are grateful to Alberto Botter for his contribution to preparing the animations reported in Figures 4 and 18 and their corresponding files in the Supplementary Materials.

Conflicts of Interest: The authors declare no conflict of interest.

#### Acronyms

| EEG    | Electroencephalogram                       |
|--------|--|
| ECG    | Electrocardiogram                          |
| EMG    | Electromyogram/electromyography            |
| sEMG   | Surface electromyogram                     |
| HDsEMG | High-density sEMG                          |
| IZ     | Innervation zone of a motor unit or muscle |
| MU     | Motor unit                                 |
| MUAP   | Motor unit action potential                |
| NMJ    | Neuromuscular junction                     |
| IED    | Interelectrode distance                    |
| EOF    | End-of-fiber                               |
| CV     | Conduction velocity of muscle fibers       |
| RMS    | Root mean square                           |
| SD     | Single differential (bipolar)              |
| DD     | Double differential                        |
| NDD    | Normal DD or Laplacian                     |
| VL     | Vastus lateralis                           |
| VM     | Vastus medialis                            |
|        |  |

#### References

- 1. Lustenberger, C.; Huber, R. High density electroencephalography in sleep research: Potential, problems, future perspective. *Front. Neurol.* **2012**, *3*, 77. [CrossRef] [PubMed]
- Dössel, O.; Luongo, G.; Nagel, C.; Loewe, A. Computer Modeling of the Heart for ECG Interpretation—A Review. *Hearts* 2021, 2, 350–368. [CrossRef]
- 3. Merletti, R.; Campanini, I.; Rymer, W.Z.; Disselhorst-Klug, C. Editorial: Surface Electromyography: Barriers Limiting Widespread Use of sEMG in Clinical Assessment and Neurorehabilitation. *Front. Neurol.* **2021**, *12*, 10–13. [CrossRef] [PubMed]
- 4. Van Campenhout, A.; Bar-On, L.; Desloovere, K.; Huenaerts, C.; Molenaers, G. Motor endplate-targeted botulinum toxin injections of the gracilis muscle in children with cerebral palsy. *Dev. Med. Child Neurol.* **2015**, *57*, 476–483. [CrossRef] [PubMed]
- Chandra, S.; Afsharipour, B.; Rymer, W.Z.; Suresh, N.L. Precise quantification of the time course of voluntary activation capacity following Botulinum toxin injections in the biceps brachii muscles of chronic stroke survivors. J. Neuroeng. Rehabil. 2020, 17, 102. [CrossRef] [PubMed]

- Rojas-Martínez, M.; Alonso, J.F.; Jordanić, M.; Mañanas, M.Á.; Chaler, J. Analysis of muscle load-sharing in patients with lateral epicondylitis during endurance isokinetic contractions using non-linear prediction. *Front. Physiol.* 2019, 10, 1185. [CrossRef]
- Zwarts, M.J.; Stegeman, D.F. Multichannel surface EMG: Basic aspects and clinical utility. *Muscle Nerve* 2003, 28, 1–17. [CrossRef]
  Drost, G.; Verrips, A.; van Engelen, B.G.M.; Stegeman, D.F.; Zwarts, M.J. Involuntary painful muscle contractions in Satoyoshi syndrome: A surface electromyographic study. *Mov. Disord.* 2006, 21, 2015–2018. [CrossRef]
- Drost, G.; Stegeman, D.F.; van Engelen, B.G.M.; Zwarts, M.J. Clinical applications of high-density surface EMG: A systematic review. J. Electromyogr. Kinesiol. 2006, 16, 586–602. [CrossRef]
- Minetto, M.A.; Lanfranco, F.; Botter, A.; Motta, G.; Mengozzi, G.; Giordano, R.; Picu, A.; Ghigo, E.; Arvat, E. Do muscle fiber conduction slowing and decreased levels of circulating muscle proteins represent sensitive markers of steroid myopathy? A pilot study in Cushing's disease. *Eur. J. Endocrinol.* 2011, 164, 985–993. [CrossRef]
- Mottram, C.J.; Heckman, C.J.; Powers, R.K.; Rymer, W.Z.; Suresh, N.L. Disturbances of motor unit rate modulation are prevalent in muscles of spastic-paretic stroke survivors. J. Neurophysiol. 2014, 111, 2017–2028. [CrossRef] [PubMed]
- 12. Campanini, I.; Merlo, A.; Farina, D. Motor unit discharge pattern and conduction velocity in patients with upper motor neuron syndrome. *J. Electromyogr. Kinesiol.* 2009, 19, 22–29. [CrossRef] [PubMed]
- Merlo, A.; Montecchi, M.G.; Lombardi, F.; Vata, X.; Musi, A.; Lusuardi, M.; Merletti, R.; Campanini, I. Monitoring Involuntary Muscle Activity in Acute Patients with Upper Motor Neuron Lesion by Wearable Sensors: A Feasibility Study. *Sensors* 2021, 21, 3120. [CrossRef] [PubMed]
- Minetto, M.; Holobar, A.; Botter, A.; Ravenni, R.; Farina, D. Mechanisms of cramp contractions: Peripheral or central generation? J. Physiol. 2011, 589, 5759–5773. [CrossRef]
- 15. Russo, A.; Aranceta-Garza, A.; D'Emanuele, S.; Serafino, F.; Merletti, R. HDSEMG activity of the lumbar erector spinae in violin players comparison of two chairs. *Med. Probl. Perform. Art.* **2019**, *34*, 205–214. [CrossRef]
- Sarcher, A.; Brochard, S.; Hug, F.; Letellier, G.; Raison, M.; Perrouin-Verbe, B.; Sangeux, M.; Gross, R. Patterns of upper limb muscle activation in children with unilateral spastic cerebral palsy: Variability and detection of deviations. *Clin. Biomech.* 2018, 59, 85–93. [CrossRef]
- Merletti, R.; Disselhorst-Klug, C.; Rymer, W.Z.; Campanini, I. Surface Electromyography: Barriers Limiting Widespread Use of sEMG in Clinical Assessment and Neurorehabilitation; Frontiers Research Topics; Frontiers Media SA: Lausanne, Switzerland, 2021; Volume 12, ISBN 9782889666164.
- 18. Campanini, I.; Disselhorst-Klug, C.; Rymer, W.Z.; Merletti, R. Surface EMG in Clinical Assessment and Neurorehabilitation: Barriers Limiting Its Use. *Front. Neurol.* **2020**, *11*, 934. [CrossRef]
- 19. Merletti, R.; Muceli, S. Tutorial. Surface EMG detection in space and time: Best practices. J. Electromyogr. Kinesiol. 2019, 49, 102363. [CrossRef]
- Merletti, R.; Cerone, G.L. Tutorial. Surface EMG detection, conditioning and pre-processing: Best practices. J. Electromyogr. Kinesiol. 2020, 54, 102440. [CrossRef]
- 21. Heckman, C.J.; Enoka, R.M. Motor Unit. In Comprehensive Physiology; J. Wiley & Sons: Hoboken, NJ, USA, 2012; Volume 2, pp. 2629–2682.
- 22. Barbero, M.; Merletti, R.; Rainoldi, A. Atlas of Muscle Innervation Zones; Springer: Milan, Italia, 2012; ISBN 978-88-470-2462-5.
- Holobar, A.; Zazula, D. Correlation-based decomposition of surface electromyograms at low contraction forces. *Med. Biol. Eng. Comput.* 2004, 42, 487–495. [CrossRef]
- 24. Merletti, R.; Roy, S.H.; Kupa, E.; Roatta, S.; Granata, A. Modeling of surface myoelectric signals—Part II: Model-based signal interpretation. *IEEE Trans. Biomed. Eng.* **1999**, *46*, 821–829. [CrossRef] [PubMed]
- 25. Merletti, R.; Lo Conte, L.; Avignone, E.; Guglielminotti, P. Modeling of surface myoelectric signals—Part I: Model implementation. *IEEE Trans. Biomed. Eng.* **1999**, *46*, 810–820. [CrossRef] [PubMed]
- 26. Disselhorst-Klug, C.; Silny, J.; Rau, G. Estimation of the relationship between the noninvasively detected activity of single motor units and their characteristic pathological changes by modelling. *J. Electromyogr. Kinesiol.* **1998**, *8*, 323–335. [CrossRef]
- 27. Farina, D.; Mesin, L.; Martina, S.; Merletti, R. A Surface EMG Generation Model with Multilayer Cylindrical Description of the Volume Conductor. *IEEE Trans. Biomed. Eng.* **2004**, *51*, 415–426. [CrossRef]
- 28. Mesin, L.; Farina, D. Simulation of Surface EMG Signals Generated by Muscle Tissues with Inhomogeneity Due to Fiber Pinnation. *IEEE Trans. Biomed. Eng.* **2004**, *51*, 1521–1529. [CrossRef]
- 29. Vieira, T.M.; Botter, A. The Accurate Assessment of Muscle Excitation Requires the Detection of Multiple Surface Electromyograms. *Exerc. Sport Sci. Rev.* 2021, 49, 23–34. [CrossRef]
- Minetto, M.A.; Botter, A.; Šprager, S.; Agosti, F.; Patrizi, A.; Lanfranco, F.; Sartorio, A. Feasibility study of detecting surface electromyograms in severely obese patients. J. Electromyogr. Kinesiol. 2013, 23, 285–295. [CrossRef]
- 31. Piervirgili, G.; Petracca, F.; Merletti, R. A new method to assess skin treatments for lowering the impedance and noise of individual gelled Ag-AgCl electrodes. *Physiol. Meas.* **2014**, *35*, 2101–2118. [CrossRef]
- Hogrel, J.-Y. Clinical applications of surface electromyography in neuromuscular disorders. *Neurophysiol. Clin. Neurophysiol.* 2005, 35, 59–71. [CrossRef]
- Afsharipour, B.; Soedirdjo, S.; Merletti, R. Two-dimensional surface EMG: The effects of electrode size, interelectrode distance and image truncation. *Biomed. Signal Process. Control* 2019, 49, 298–307. [CrossRef]
- 34. Schwartz, M.W. EMG Methods for Evaluating Muscle and Nerve Function; Schwartz, M., Ed.; InTech: London, UK, 2012; Volume 68, ISBN 978-953-307-793-2.

- Merletti, R.; Farina, D. Surface Electromyography: Physiology, Engineering, and Application; Merletti, R., Farina, D., Eds.; IEEE Press: Hoboken, NJ, USA; John Wiley & Sons, Inc.: Hoboken, NJ, USA, 2016; ISBN 9781119082934.
- Besomi, M.; Hodges, P.W.; Van Dieën, J.; Carson, R.G.; Clancy, E.A.; Disselhorst-Klug, C.; Holobar, A.; Hug, F.; Kiernan, M.C.; Lowery, M.; et al. Consensus for experimental design in electromyography (CEDE) project: Electrode selection matrix. *J. Electromyogr. Kinesiol.* 2019, 48, 128–144. [CrossRef] [PubMed]
- Farina, D.; Madeleine, P.; Graven-Nielsen, T.; Merletti, R.; Arendt-Nielsen, L. Standardising surface electromyogram recordings for assessment of activity and fatigue in the human upper trapezius muscle. *Eur. J. Appl. Physiol.* 2002, *86*, 469–478. [CrossRef] [PubMed]
- Farina, D.; Merletti, R.; Indino, B.; Nazzaro, M.; Pozzo, M. Surface EMG crosstalk between knee extensor muscles: Experimental and model results. *Muscle Nerve* 2002, 26, 681–695. [CrossRef] [PubMed]
- 39. Merlo, A.; Bò, M.C.; Campanini, I. Electrode Size and Placement for Surface EMG Bipolar Detection from the Brachioradialis Muscle: A Scoping Review. *Sensors* 2021, *21*, 7322. [CrossRef] [PubMed]
- 40. Nishihara, K.; Chiba, Y.; Suzuki, Y.; Moriyama, H.; Kanemura, N.; Ito, T.; Takayanagi, K.; Gomi, T. Effect of position of electrodes relative to the innervation zone on surface EMG. *J. Med. Eng. Technol.* **2010**, *34*, 141–147. [CrossRef]
- Rodrigues, F.B.; Duarte, G.S.; Marques, R.E.; Castelão, M.; Ferreira, J.; Sampaio, C.; Moore, A.P.; Costa, J. Botulinum toxin type A therapy for cervical dystonia. *Cochrane Database Syst. Rev.* 2020, 2020, CD003633. [CrossRef]
- 42. Lapatki, B.G.; Van Dijk, J.P.; Van de Warrenburg, B.P.C.; Zwarts, M.J. Botulinum toxin has an increased effect when targeted toward the muscle's endplate zone: A high-density surface EMG guided study. *Clin. Neurophysiol.* **2011**, 122, 1611–1616. [CrossRef]
- Zoons, E.; Dijkgraaf, M.G.W.; Dijk, J.M.; van Schaik, I.N.; Tijssen, M.A. Botulinum toxin as treatment for focal dystonia: A systematic review of the pharmaco-therapeutic and pharmaco-economic value. J. Neurol. 2012, 259, 2519–2526. [CrossRef]
- 44. Sultan, A.H.; Kamm, M.A.; Bartram, C.I.; Hudson, C.N. Anal sphincter trauma during instrumental delivery. *Int. J. Gynecol. Obstet.* **1993**, 43, 263–270. [CrossRef]
- Cescon, C.; Riva, D.; Začesta, V.; Drusany-Starič, K.; Martsidis, K.; Protsepko, O.; Baessler, K.; Merletti, R. Effect of vaginal delivery on the external anal sphincter muscle innervation pattern evaluated by multichannel surface EMG: Results of the multicentre study TASI-2. *Int. Urogynecol. J.* 2014, 25, 1491–1499. [CrossRef]
- 46. Enck, P.; Franz, H.; Davico, E.; Mastrangelo, F.; Mesin, L.; Merletti, R. Repeatability of innervation zone identification in the external anal sphincter muscle. *Neurourol. Urodyn.* **2010**, *29*, 449–457. [CrossRef] [PubMed]
- Peng, Y.; He, J.; Khavari, R.; Boone, T.B.; Zhang, Y. Functional mapping of the pelvic floor and sphincter muscles from high-density surface EMG recordings. *Int. Urogynecol. J.* 2016, 27, 1689–1696. [CrossRef] [PubMed]
- 48. Ramaekers, V.; Disselhorst-Klug, C.; Schneider, J.; Silny, J.; Forst, J.; Forst, R.; Kotlarek, F.; Rau, G. Clinical Application of a Noninvasive Multi-Electrode Array EMG for the Recording of Single Motor Unit Activity. *Neuropediatrics* **1993**, *24*, 134–138. [CrossRef] [PubMed]
- 49. Disselhorst-Klug, C.; Bahm, J.; Ramaekers, V.; Trachterna, A.; Rau, G. Non-invasive approach of motor unit recording during muscle contractions in humans. *Eur. J. Appl. Physiol.* **2000**, *83*, 144–150. [CrossRef] [PubMed]
- 50. Buchthal, F.; Kamieniecka, Z. The diagnostic yield of quantified electromyography and quantified muscle biopsy in neuromuscular disorders. *Muscle Nerve* **1982**, *5*, 265–280. [CrossRef]
- Moloney, P.B.; Lefter, S.; Ryan, A.M.; Jansen, M.; Bermingham, N.; McNamara, B. The Diagnostic Yield of Electromyography at Detecting Abnormalities on Muscle Biopsy: A Single Center Experience. *Neurodiagn. J.* 2021, *61*, 86–94. [CrossRef]
- Stegeman, D.F.; Kleine, B.U.; Lapatki, B.G.; Van Dijk, J.P. High-density Surface EMG: Techniques and Applications at a Motor Unit Level. *Biocybern. Biomed. Eng.* 2012, 32, 3–27. [CrossRef]
- 53. Rasool, G.; Afsharipour, B.; Suresh, N.L.; Rymer, W.Z. Spatial Analysis of Multichannel Surface EMG in Hemiplegic Stroke. *IEEE Trans. Neural Syst. Rehabil. Eng.* 2017, 25, 1802–1811. [CrossRef]
- 54. Williams, S.E.; Koch, K.C.; Disselhorst-Klug, C. Non-invasive assessment of motor unit activation in relation to motor neuron level and lesion location in stroke and spinal muscular atrophy. *Clin. Biomech.* **2020**, *78*, 105053. [CrossRef]
- 55. Huppertz, H.J.; Disselhorst-Klug, C.; Silny, J.; Rau, G.; Heimann, G. Diagnostic yield of noninvasive high spatial resolution electromyography in neuromuscular diseases. *Muscle Nerve* **1997**, *20*, 1360–1370. [CrossRef]
- 56. Rainoldi, A.; Gazzoni, M.; Casale, R. Surface EMG signal alterations in Carpal Tunnel syndrome: A pilot study. *Eur. J. Appl. Physiol.* **2008**, *103*, 233–242. [CrossRef] [PubMed]
- Wren, T.A.L.; Tucker, C.A.; Rethlefsen, S.A.; Gorton, G.E.; Õunpuu, S. Clinical efficacy of instrumented gait analysis: Systematic review 2020 update. *Gait Posture* 2020, 80, 274–279. [CrossRef] [PubMed]
- Campanini, I.; Cosma, M.; Manca, M.; Merlo, A. Added Value of Dynamic EMG in the Assessment of the Equinus and the Equinovarus Foot Deviation in Stroke Patients and Barriers Limiting Its Usage. *Front. Neurol.* 2020, 11, 583399. [CrossRef] [PubMed]
- 59. Merlo, A.; Campanini, I. Impact of instrumental analysis of stiff knee gait on treatment appropriateness and associated costs in stroke patients. *Gait Posture* **2019**, 72, 195–201. [CrossRef]
- 60. Campanini, I.; Merlo, A.; Damiano, B. A method to differentiate the causes of stiff-knee gait in stroke patients. *Gait Posture* **2013**, 38, 165–169. [CrossRef]
- 61. Wren, T.A.L.; Chou, L.S.; Dreher, T. Gait and posture virtual special Issue "clinical impact of instrumented motion analysis." *Gait Posture* **2020**, *82*, 108–109. [CrossRef]
- Wren, T.A.L.; Gorton, G.E.; Õunpuu, S.; Tucker, C.A. Efficacy of clinical gait analysis: A systematic review. *Gait Posture* 2011, 34, 149–153. [CrossRef]

- Ferrarin, M.; Rabuffetti, M.; Bacchini, M.; Casiraghi, A.; Castagna, A.; Pizzi, A.; Montesano, A. Does gait analysis change clinical decision-making in poststroke patients? Results from a pragmatic prospective observational study. *Eur. J. Phys. Rehabil. Med.* 2015, *51*, 171–184.
- 64. Mazzoli, D.; Giannotti, E.; Manca, M.; Longhi, M.; Prati, P.; Cosma, M.; Ferraresi, G.; Morelli, M.; Zerbinati, P.; Masiero, S.; et al. Electromyographic activity of the vastus intermedius muscle in patients with stiff-knee gait after stroke. A retrospective observational study. *Gait Posture* 2018, *60*, 273–278. [CrossRef]
- 65. Blanc, Y. EMG timing errors of pathologic gait. In *Proceedings of the First General SENIAM* (Surface EMG for Non- Invasive Assessment of Muscles) Workshop; Hermens, H.J., Ed.; RRD Publisher: Enschede, The Netherlands, 1996; pp. 183–185.
- Campanini, I.; Merlo, A.; Degola, P.; Merletti, R.; Vezzosi, G.; Farina, D. Effect of electrode location on EMG signal envelope in leg muscles during gait. J. Electromyogr. Kinesiol. 2007, 17, 515–526. [CrossRef]
- Benedetti, M.G.; Beghi, E.; De Tanti, A.; Cappozzo, A.; Basaglia, N.; Cutti, A.G.; Cereatti, A.; Stagni, R.; Verdini, F.; Manca, M.; et al. SIAMOC position paper on gait analysis in clinical practice: General requirements, methods and appropriateness. Results of an Italian consensus conference. *Gait Posture* 2017, *58*, 252–260. [CrossRef] [PubMed]
- 68. Hermens, H.; Freriks, B.; Merletti, R.; Stegeman, D.; Blok, J.; Rau, G.; Disselhorst-Klug, C.; Hagg, G. European Recommendations for Surface Electromyography; RRD Publisher: Enschede, The Netherlands, 1999; ISBN 90-75452-15-2.
- Blanc, Y. Electrode Placement for surface EMG. In AIM Project A2002 CAMARC-II (Computer Aided Movement Analysis in a Rehabilitation Context-II): Functional Evaluation Protocolos for Europe-Wide Network of Clinical Centres; RRD Publisher: Enschede, The Netherlands, 1994; pp. 25–30.
- 70. Blanc, Y.; Dimanico, U. Electrode Placement in Surface Electromyography (sEMG) "Minimal Crosstalk Area" (MCA). *Open Rehabil. J.* **2014**, *3*, 110–126. [CrossRef]
- 71. Blumenstein, R.; Basmajian, J. Electrode Placement in EMG Biofeedback; Williams & Wilkins: Baltimore, MA, USA, 1980.
- Merlo, A.; Campanini, I. Technical Aspects of Surface Electromyography for Clinicians. *Open Rehabil. J.* 2010, *3*, 98–109. [CrossRef]
  Merlo, A.; Campanini, I. Applications in movement and gait analysis. In *Surface Electromyography: Physiology, Engineering, and Applications*; John Wiley & Sons, Inc.: Hoboken, NJ, USA, 2016; pp. 440–459. ISBN 978-1-118-98702-5.
- 74. Shewman, T.; Konrad, P. Clinical Sequence Assessments and sEMG feedback. A Beginner's Guide; Noraxon Inc.: Scottsdale, AZ, USA, 2011; ISBN 0977162249.
- 75. Kofler, M.; Kreczy, A.; Gschwendtner, A. "Occupational backache"—Surface electromyography demonstrates the advantage of an ergonomic versus a standard microscope workstation. *Eur. J. Appl. Physiol.* **2002**, *86*, 492–497. [CrossRef] [PubMed]
- Felici, F. Neuromuscular responses to exercise investigated through surface EMG. J. Electromyogr. Kinesiol. 2006, 16, 578–585. [CrossRef] [PubMed]
- 77. Felici, F.; Del Vecchio, A. Surface Electromyography: What Limits Its Use in Exercise and Sport Physiology? *Front. Neurol.* **2020**, 11, 578504. [CrossRef]
- 78. Martin, B.J.; Acosta-Sojo, Y. sEMG: A Window into Muscle Work, but Not Easy to Teach and Delicate to Practice—A Perspective on the Difficult Path to a Clinical Tool. *Front. Neurol.* **2021**, *11*, 588451. [CrossRef]
- 79. Soderberg, G.L. Selected Topics in Surface Electromyography for Use in the Occupational Setting: Expert Perspectives; US Department of Health and Human Services: Washington, DC, USA, 1992; Volume 1.
- 80. Kumar, S.; Mital, A. *Electromyography in Ergonomics*, 2nd ed.; Shrawan, K., Anil, M., Eds.; Routledge: England, UK, 2017; ISBN 9780203758670.
- 81. Marras, W.S. Overview of Electromyography in Ergonomics. Proc. Hum. Factors Ergon. Soc. Annu. Meet. 2000, 44, 5-534-5-536. [CrossRef]
- 82. Gazzoni, M.; Afsharipour, B.; Merletti, R. Surface EMG in Ergonomics and Occupational Medicine. In *Surface Electromyography: Physiology, Engineering, and Applications*; John Wiley & Sons, Inc.: Hoboken, NJ, USA, 2016; pp. 361–391.
- 83. Varrecchia, T.; Ranavolo, A.; Conforto, S.; De Nunzio, A.M.; Arvanitidis, M.; Draicchio, F.; Falla, D. Bipolar versus high-density surface electromyography for evaluating risk in fatiguing frequency-dependent lifting activities. *Appl. Ergon.* **2021**, *95*, 103456. [CrossRef]
- 84. Clarys, J.P.; Cabri, J. Electromyography and the study of sports movements: A review. J. Sports Sci. 1993, 11, 379–448. [CrossRef]
- 85. Türker, H.; Sözen, H. Surface Electromyography in Sport and Exercise. Chapter 9; Turker, D.H., Ed.; InTech: London, UK, 2013; ISBN 978-953-51-1118-4.
- 86. Rainoldi, A.; Gazzoni, M.; Melchiorri, G. Differences in myoelectric manifestations of fatigue in sprinters and long distance runners. *Physiol. Meas.* **2008**, *29*, 331–340. [CrossRef] [PubMed]
- 87. Farago, E.; Macisaac, D.; Suk, M.; Chan, A.D.C. A Review of Techniques for Surface Electromyography Signal Quality Analysis. *IEEE Rev. Biomed. Eng.* **2022**. *in press*. [CrossRef] [PubMed]
- Ivanenko, Y.P.; D'avella, A.; Lacquaniti, F. Muscle Coordination, Motor Synergies, and Primitives from Surface EMG. In Surface Electromyography: Physiology, Engineering, and Applications; John Wiley & Sons, Inc.: Hoboken, NJ, USA, 2016; pp. 158–179. ISBN 9781119082934.
- 89. Besomi, M.; Hodges, P.W.; Clancy, E.A.; Van Dieën, J.; Hug, F.; Lowery, M.; Merletti, R.; Søgaard, K.; Wrigley, T.; Besier, T.; et al. Consensus for experimental design in electromyography (CEDE) project: Amplitude normalization matrix. *J. Electromyogr. Kinesiol.* **2020**, *53*, 102438. [CrossRef] [PubMed]
- McManus, L.; Lowery, M.; Merletti, R.; Søgaard, K.; Besomi, M.; Clancy, E.A.; van Dieën, J.H.; Hug, F.; Wrigley, T.; Besier, T.; et al. Consensus for experimental design in electromyography (CEDE) project: Terminology matrix. J. Electromyogr. Kinesiol. 2021, 59, 102565. [CrossRef]

- 91. Rogers, E.M. Diffusion of Innovation; The Free Press: New York, NY, USA, 2003; ISBN 9780743258234.
- 92. Jette, A.M. Overcoming Ignorance and Ineptitude in 21st Century Rehabilitation. Phys. Ther. 2017, 97, 497–498. [CrossRef]
- Hermens, H.J.; Freriks, B.; Disselhorst-Klug, C.; Rau, G. Development of recommendations for SEMG sensors and sensor placement procedures. J. Electromyogr. Kinesiol. 2000, 10, 361–374. [CrossRef]
- 94. Manca, A.; Cereatti, A.; Bar-On, L.; Botter, A.; Croce, U.D.; Knaflitz, M.; Maffiuletti, N.A.; Mazzoli, D.; Merlo, A.; Roatta, S.; et al. A Survey on the Use and Barriers of Surface Electromyography in Neurorehabilitation. *Front. Neurol.* **2020**, *11*, 573616. [CrossRef]
- 95. Jette, A.M. Moving research from the bedside into practice. *Phys. Ther.* **2016**, *96*, 594–596. [CrossRef]
- 96. Gawande, A. Slow Ideas Some innovations spread fast. How do you speed the ones that don't? New Yorker 2013, 29, 36-45.